Author’s response to reviews

Title: In search of quality evidence for lifestyle management and glycemic control in children and adolescents with type 2 diabetes: A systematic review

Authors:

Steve T Johnson (sjohnson@athabascau.ca)
Amanda S Newton (mandi.newton@ualberta.ca)
Meera Chopra (mchopra@mtsinai.on.ca)
Jeanette Buckingham (jeanette.buckingham@ualberta.ca)
Terry T-K Huang (huangter@mail.nih.gov)
Paul W Franks (paul.franks@medicin.umu.se)
Mary M Jetha (mary.jetha@albertahealthservices.ca)
Geoff DC Ball (gdball@ualberta.ca)

Version: 8 Date: 20 December 2010

Author’s response to reviews:

A revised version of our manuscript that includes major revisions is attached.